Author's response to reviews

Title: Homeopathy for Depression - DEP-HOM: study protocol for a randomized, partially double-blind, placebo controlled, four armed study

Authors:

Ubiratan C Adler (ubiratan.adler@charite.de)
Stephanie Krüger (stephanie.krueger@charite.de)
Michael Teut (michael.teut@charite.de)
Rainer Lüdtke (r.luedtke@carstens-stiftung.de)
Iris Bartsch (iris.bartsch@charite.de)
Lena Schützler (lena.schuetzler@charite.de)
Friederike Melcher (friederike.melcher@charite.de)
Stephan N Willich (stefan.willich@charite.de)
Klaus Linde (klaus.linde@lrz.tum.de)
Claudia M Witt (claudia.witt@charite.de)

Version: 8 Date: 3 February 2011

Author's response to reviews:

Dr. Doug Altman, Dr. Curt Furberg, Dr. Jeremy Grimshaw and Dr. Peter Rothwell
Editors-in-Chief
Trials Journal

MS: 1103141824456928

Homeopathy for Depression - DEP-HOM: study protocol for a randomized, partially double-blind, placebo controlled, four armed study

Berlin, February 3rd, 2011

Dear Editor

We have made the following major changes, according to your suggestion: a Discussion section has been included in the Abstract. The box was formatted in a list and added as an Appendix before the Acknowledgements section. These changes are highlighted in red.

Please note that Figure 1 is already in the ms.

With best wishes,

Ubiratan C. Adler